
Access

Stacking:
○ Don't forget to cut out numbers and details when giving report. Try to talk in 
generalities and not mention numbers or specific device names unless there is a 
major problem.

○ Protons on target ran 7.5e12 most of the weekend 11 turns.  Dropped to 7.4e12 by 
the end of the weekend.

○ Best stacking hour on Friday was 24.5mA, close to best stacking conditions.   
Stacking fell off a few percent over the weekend to both upstream performance 
degradation as well as normal target performance dropping.  

○ Paul's pbar plots and numbers did not update on Sunday.  Paul will re-run these 
today.  

○ Cannont connect to Java to run Java performance tools and plot this morning.   Get 
an error connecting to DSE06 - connection refused.

○ We set a record one week of stacking with 2,863mA.

○ Friday

� D:Q724 tripped during the evening shift.
○ Saturday

� 15:10:20- The Pbar overthruster was restarted from the most recent "I Like 
It" file because I:VT703 was exceeding its maximum limit.  Lowered the flattop 
value for I:LAM52 on the Pbar cycles to bring down the current in I:VT703.

○ Sunday

� 20:53:50- Rebooted PBCOOL VME at the request of DVM. P34 was getting 
MOOC_NO_SUCH_OID errors and settings to the core momentum systems were 
not taking. After the reboot, things worked normally again.

� After the reboot, D59 still bombs when trying to unbypass D:PV3AU1.

○ Monday

� Tony made a target move just prior to the 9am meeting.

2008-02-11 Monday Morning Pbar Notes

Sunday, February 10, 2008
6:01 PM
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Transfers:
○ MI collimators were pulled all the way out for this weekend. 
○ Transfers for the most part good.  There was not a noticeable gain by pulling the 

collimators out.
○ Some sub-par transfers on Saturday and Sunday evening.  

� Saturday
□ 7105 - 88.2%, Saturday, 2/9 12:26:16pm
□ 7106 - 87.1%, Saturday, 2/9 1:49:46pm
□ Emittance management.

� Sunday
□ 7122 - 88.1%, Sunday, 2/10 4:24:51pm
□ 7123 89.7%
□ 7124 - 87.9%
□ 7125 - 86.7%
□ 7136 - 89.2% - Sunday, 2/10 11pm

○ Problems

� 7109 - missing IBEAM2
� 7098 - I:BEAMS large nubmer
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� 7098 - I:BEAMS large nubmer

Studies:

Requests:
○ Gain Ramping Studies

� "Completely parasitic"
� Take over SA#1 (bunch rotation), SA#2 (Accumulator Longitudinal profile), and  

the D/A VSA (Accumulator to Debuncher alignment).  
� D/A VSA also takes away the Stacktail Monitor VSA during stacking.   

Unstacking mode in the Accumulator VSA should be ok. 

Other Notes:
○ Paul's Numbers

� Saturday
□ Most in an hour: 24.64 mA at Fri Feb 08 21:56:00 CST 2008 
□ Best: 25.19 mA on 30-Jan-08 
□ Average Production 17.16 e-6/proton Best: 25.41 e-6/proton on 

01/30/2008 
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□
01/30/2008 

□ Average Protons on Target 6.70 e12 Best: 8.77 e12 on 07/24/2007 
□ Largest Stack 59.59 mA Best: 271.01 mA on 11/14/2007 

� Sunday
□ Didn't update

� Monday
○ Most in an hour: 23.68 mA at Sun Feb 10 19:10:31 CST 2008 
○ Best: 25.19 mA on 30-Jan-08 
○ Average Production 17.90 e-6/proton Best: 25.41 e-6/proton on 

01/30/2008 
○ Average Protons on Target 6.39 e12 Best: 8.77 e12 on 07/24/2007 
○ Largest Stack 49.12 mA Best: 271.01 mA on 11/14/2007 

○ Al's Numbers

� Cannot connect to DSE06 - was able to update later
Stacking
Pbars stacked: 482.14 E10
Time stacking: 22.46 Hr
Average stacking rate: 21.46 E10/Hr

Uptime
Number of pulses while in stacking mode: 36255
Number of pulses with beam: 34255
Fraction of up pulses was: 94.48%

The uptime's effect on the stacking numbers
Corrected time stacking: 21.22 Hr
Possible average stacking rate: 22.72 E10/Hr
Could have stacked: 510.29 E10/Hr

Recycler Transfers
Pbars sent to the Recycler: 474.53 E10
Number of transfers : 42
Number of transfer sets: 14
Average Number of transfer per set: 3.00
Time taken to shoot: 01.54 Hr
Time per set of transfers: 06.58 min
Transfer efficiency: 90.68%

Other Info
Average POT : 7.42 E12
Average production: 18.96 pbars/E6 protons

○ Runing without a DRF cavity.  Return instructions.
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Pasted from <http://www-bd.fnal.gov/cgi-mach/machlog.pl?nb=pbar08&action=view&page=-800
&button=yes&invert=yes> 

○ Poor transfers on Saturday evening due to emittance management issues
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